Author's response to reviews

Title: Intervention to reduce excessive alcohol consumption & improve co-morbid outcomes in hypertensive or depressed primary care patients: two parallel randomised feasibility trials

Authors:

Graeme B Wilson (graeme.wilson@ncl.ac.uk)
Catherine Wray (ciwray64@gmail.com)
Ruth McGovern (r.mcGovern@newcastle.ac.uk)
Dorothy Newbury-Birch (dorothy.newbury-birch@newcastle.ac.uk)
Elaine McColl (Elaine.Mccoll@newcastle.ac.uk)
Ann Crosland (ann.crosland@sunderland.ac.uk)
Chris Speed (chris.speed@newcastle.ac.uk)
Paul Cassidy (Paul.cassidy@gp-a85023.nhs.uk)
Dave Tomson (D.P.C.Tomson@newcastle.ac.uk)
Shona Haining (Shona.Haining@northoftyne.nhs.uk)
Denise Howel (denise.howel@newcastle.ac.uk)
Eileen FS Kaner (eileen.kaner@newcastle.ac.uk)

Version: 4 Date: 23 May 2014

Author's response to reviews:

Hello,

Please find attached a manuscript revised as per the editorial requests in an email from Jessie McGavin on 21st May 2014. The specific changes made are:

1. We have added the date the study was registered next to the trial registration number at the end of the Abstract.

2. A list of abbreviations used and their meanings has been added after the Conclusion; the abbreviation ‘UK’ for United Kingdom has been replaced with the full spelling throughout.

3. Details have been added to the Authors’ Contributions section to specify the contribution of author Chris Speed.

In the attached file, these changes have been added to those tracked in the revision submitted on 27th April 2014. Do please let me know if you need a version with only these most recent changes tracked, or if any further amendments are required. Thank you for your help,

With best wishes

Graeme Wilson